Author's response to reviews

Title: The use of conspicuity aids by cyclists and risk of crashes involving other road users: a protocol for a population based case-control study

Authors:

Philip D Miller (mcxpdm@nottingham.ac.uk)
Denise Kendrick (denise.kendrick@nottingham.ac.uk)
Carol Coupland (carol.coupland@nottingham.ac.uk)
Frank Coffey (frank.coffey@nuh.nhs.uk)

Version: 4 Date: 18 January 2010

Author's response to reviews:

1) Names reconciled with the submission system
2) email Addresses moved back to title page
3) Abbreviations given as a paragraph
4) Font changed to Arial